**Author's response to reviews**

**Title:** Determinants of participation in a longitudinal two-stage study of the health consequences of the Chornobyl nuclear power plant accident

**Authors:**

Lin T Guey (ltung@cnio.es)
Evelyn J Bromet (ebromet@notes.cc.sunysb.edu)
Semyon Gluzman (upa2@i.com.ua)
Victoria Zakhozha (victoria@kiis.com.ua)
Vlodomyr Paniotto (paniotto@kmis.kiev.ua)

**Version:** 5  **Date:** 25 April 2008

**Author's response to reviews:**

We have made the following changes to the manuscript:

1) "Guey et al" has been removed from the header of each page

2) A figure legend has been added to Figure 1

3) Table 4 has been reformatted so that it fits on an A4 page

4) the authors list has been corrected

Thank you for pointing out these formatting details,

Lin